Author's response to reviews

Title: Decrease social inequalities return-to-work: development and design of a randomised controlled trial among women with breast cancer.

Authors:

Clémence Vidor (clemence.vidor@univ-lille3.fr)
Ariane Leroyer (ariane.leroyer@univ-lille2.fr)
Véronique Christophe (veronique.christophe@univ-lille3.fr)
Mélanie Seillier (melanie.seillier@carsat-nordpicardie.fr)
Jérome Foncel (jerome.foncel@univ-lille3.fr)
Jacques Bonneterre (j-bonneterre@o-lambret.fr)
Sophie Fantoni (fanquin@wanadoo.fr)

Version: 6 Date: 19 March 2014

Author's response to reviews:

- Files 1 to 4 have been removed
- The name and reference of the Ethics Committee were added page 12
- The title page was added